Author’s response to reviews

Title: Application of Latent Semantic Analysis for Open-Ended Responses in a Large, Epidemiologic Study

Authors:

Travis D Leleu (travis.leleu@med.navy.mil)
Isabel G Jacobson (isabel.jacobson@med.navy.mil)
Cynthia A LeardMann (cynthia.leardmann@med.navy.mil)
Besa Smith (besa.smith@med.navy.mil)
Peter W Foltz (pfoltz@pearsonkt.com)
Paul J Amoroso (pjamoroso@aol.com)
Marcia A Derr (marcia.derr@pearsonkt.com)
Margaret AK Ryan (margaret.ryan@med.navy.mil)
Tyler C Smith (tsmith@nu.edu)
for the Millennium Cohort Study Team (info@millenniumcohort.org)

Version: 5 Date: 30 September 2011

Author’s response to reviews: see over
September 30, 2011

Adrian Aldcroft, BA  
Executive Editor  
BMC Medical Research Methodology

Dear Editor Aldcroft,

Enclosed please find our revised manuscript, “Application of Latent Semantic Analysis for Open-Ended Responses in a Large, Epidemiologic Study.”

The authors have revised the manuscript as recommended below:

Major revisions (we require the author to make these changes)

Abstract: please ensure that the Abstract is present in the manuscript and matches exactly what is in the submission system. Currently they do not match.  
**We have made sure the abstract from the Word document matches the abstract in the online submission system.**

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.  
**Authors have reviewed and edited the manuscript for typographical errors.**

Sincerely,

Isabel Jacobson, MPH